Author’s response to reviews

Title: Requirements of Health Policy And Services journals for authors to disclose financial and non-financial conflicts of interest: a cross-sectional study

Authors:

Assem Khamis (assemabdelmoaty@outlook.com)
Maram Hakoum (maram.hakoum@gmail.com)
Lama Bou-Karroum (lb25@aub.edu.lb)
Joseph Habib (joseph.habib12@live.com)
Ahmed Ali (aka36@mail.aub.edu)
Gordon Guyatt (guyatt@mcmaster.ca)
Fadi El-Jardali (fe08@aub.edu.lb)
Elie Akl (ea32@aub.edu.lb)

Version: 3 Date: 23 Aug 2017

Author’s response to reviews:

Clinical Research Institute
American University of Beirut, Lebanon

August 23, 2017

Rosanna Gonzalez-Mcquire, Editor

Subject: “Requirements of Health Policy and Services journals for authors to disclose financial and non-financial conflicts of interest: a cross-sectional study”
Dear Dr. Gonzalez-Mcquire,

We thank you for the opportunity to revise our manuscript for your consideration for publication in Health Research Policy and Systems.

Please find on the following pages the minor correction requested. We will be happy to address any further comments or suggestions you or the reviewers might have.

With kind regards,

Elie A. Akl, M.D., M.P.H., Ph.D.
Associate Professor of Medicine
American University of Beirut, Lebanon

Please note: reviewers’ comments are in bold font, our responses are in regular font and, and extracts from the text are in italic and placed within “quotation marks” with any changes underlined.

Chief Editors

Comment 1:
In Table 6 heading, please replace ‘should’ with ‘could’, which presumably as an oversight still says ‘should’ instead of ‘could’.

Response 1:
“Table 6: suggestions for items that journals could include in their policies for authors’ disclosure of COI”